Author's response to reviews

Title: Haemodynamics and oxygenation improvement induced by High Frequency Percussive Ventilation in a hypoxic cardiac surgery patient: a case report

Authors:

alessandro forti (alefortidoc@me.com)
valeria salandin (vsalandin@ulss.tv.it)
paolo zanatta (pzanatta@mac.com)
bruno persi (bpersi@ulss.tv.it)
carlo sorbara (csorbara@ulss.tv.it)

Version: 8 Date: 14 September 2010

Author's response to reviews:

Dear Editor, I have correct the major and minor revision: MAJOR: 1. The title on the Manuscript is the same on the submitted version; 2. References are cited in square brackets; 3. Citation of figure 3 now is correct ("Figure 3 shows PaO2 increasing after HFPV"); MINOR: 1. Author list is now correct, 2. The consent section is after Competing Interests; 3. I add the corresponding author at the end of the title page; 4. I have remove from the tables the vertical lines; 5. the references list are now without the term "et al."; 6. The figures are now cropped as closely as possible; 7. I have add in abbreviations the term HFV (high frequency ventilation).

Thank You

Kind regards

Alessandro Forti